Author’s response to reviews

Title: Changes in CT morphology can be an independent response marker for patients receiving regorafenib for colorectal liver metastases: retrospective pilot study

Authors:
Yukinori Ozaki (1755ozaki@gmail.com)
Junichi Shindoh (shindou-tyky@umin.ac.jp)
Wataru Gono (watapi-tyky@umin.net)
Yujiro Nishioka (yutan21woodstock@gmail.com)
Chihiko Kondoh (ckondoh@toranomon.gr.jp)
Yuko Tanabe (ytanabe@toranomon.gr.jp)
Shuichiro Matoba (s-matoba@ac.cyberhome.ne.jp)
Hiroya Kuroyanagi (kuroyanagi@toranomon.gr.jp)
Masaji Hashimoto (masaji.hashimoto@gmail.com)
Toshimi Takano (takano@toranomon.gr.jp)

Version: 4 Date: 26 Jan 2018

Author’s response to reviews:

Editor Comments:

Thank you very much for getting back to us about re-use and permission for the figure. In the notes of the license it mentions that "with permission of Springer" should be stated, so we ask that this is added to the additional file legend where you also cite the publication in question [reference 8].

→We have added "with permission of Springer" to the figure legend.

Could we also ask that you clarify the Ethics and consent to participate to section relating to the 'website'. Please provide the website URL, or consider removing reference to the website and including any brief relevant additional details relating to consent in this section instead. As a very minor point, please change the heading of this section from Ethics approval to 'Ethics and consent to participate'.
We have added the website URL and change the heading title to "Ethics and consent to participate"

Thank you for your supports.